Author’s response to reviews

Title: A case of autism with an interstitial deletion on 4q leading to hemizygosity for genes encoding for glutamine and glycine neurotransmitter receptor sub-units (AMPA 2, GLRA3, GLRB) and neuropeptide receptors NPY1R, NPY5R

Authors:

Subhadra Ramanathan (subhs@yahoo.com)
Abigail Woodroffe (a_woodroffe@hotmail.com)
Pamela L Flodman (pflodman@uci.edu)
Lee Z Mays (leezmays@hotmail.com)
Mona Hanouni (mhanouni@uci.edu)
Charlotte B Modahl (cmodahl@uci.edu)
Robin Steinberg-Epstein (rsteinbe@uci.edu)
Maureen E Bocian (mebocian@uci.edu)
M. Anne Spence (maspence@uci.edu)
Prof Moyra Smith (dmsmith@uci.edu)

Version: 3 Date: 14 Apr 2004

April 14, 2004

Dear BMC Editors:

According to the formatting changes recently requested, we made the following revisions. We altered the title page to include the email addresses for all authors. We also indicated each author’s corresponding affiliation with the use of superscript numbers. Next, we ensured that the Abstract appears on the second page of the manuscript and not on the title page. In addition, we removed all unnecessary capitalization in the section headings, ie. we changed "Case Presentation" to read "Case presentation." Finally, we placed the details and description of the molecular studies under the Case presentation heading.

We hope that our formatting revisions were made to your satisfaction. Thank you for your time.

Sincerely,
Moyra Smith, MD, PhD